Author's response to reviews

Title: Nursing home care for people with dementia and residents' quality of life, quality of care and staff well-being: Design of the Living Arrangements for people with Dementia (LAD) - study

Authors:

Bernadette M. Willemse (bwillemse@trimbos.nl)
Dieneke Smit (dsmit@trimbos.nl)
Jacomine de Lange (jlange@trimbos.nl)
Anne Margriet Pot (ampot@trimbos.nl)

Version: 3 Date: 10 March 2011

Author's response to reviews: see over
Dear Sir/Madam,

We hereby submit our revised manuscript entitled 'Nursing home care for people with dementia and residents' quality of life, quality of care, staff well-being: Design of the Living Arrangements for people with Dementia (LAD) – study' (first submitted at 25-01-2011 and revised 02-03-2011) for publication in *BMC Geriatrics*.

We were very pleased to hear that our manuscript has been accepted for publication in *BMC Geriatrics*. The editorial team asked us to make some final revisions in the section on the authors' contributions and the heading of the methods section. We have made these changes and took the opportunity to make some final corrections.

Also on behalf of the co-authors,

Yours sincerely,

Bernadette Willemse, MSc